Author's response to reviews

Title: Association of HIV infection with distribution and viral load of HPV types in Kenya: a survey with 820 female sex workers

Authors:

Stanley MF Luchters (stanley.luchters@ugent.be)
Davy Vanden Broeck (davy.vandenbroeck@ugent.be)
Matthew F Chersich (chersich@doctors.org.uk)
Annalene Nel (anel@ipm-microbicides.org)
Wim Delva (wim.delva@ugent.be)
Kishor Mandaliya (kmandaliya@gmail.com)
Christophe Depuydt (christophe.depuydt@riatol.be)
Patricia Claeys (Patricia.Claeys@UGent.Be)
John-Paul Bogers (john-paul.bogers@ua.ac.be)
Marleen Temmerman (marleen.temmerman@ugent.be)

Version: 3 Date: 23 January 2010

Author's response to reviews:

Author list: Stanley MF Luchters, Davy Vanden Broeck, Matthew F Chersich, Annalene Nel, Wim Delva, Kishor Mandaliya, Christophe Depuydt, Patricia Claeys, John-Paul Bogers and Marleen Temmerman
Title: Association of HIV infection with distribution and viral load of HPV types in Kenya: a survey with 820 female sex workers
Journal: BMC Infectious Diseases
MS ID: 1063823391308943

Dear editor,

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)
Response: This has been checked and amended accordingly

Conclusions - Please insert a Conclusions section after the Discussion.
Response: this has been done

Tables: please note that we are unable to display text vertically within the cells: please re-layout your table without these vertical elements.
Response: tables amended accordingly

Minor revisions (we can make these changes for you, although it will speed up
publication of your manuscript if you do them while making the major changes above)
Word count: please remove these
Response: this has been done
Key words: please remove these
Response: this has been done
References: please provide the full citation details for reference 2. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references.
Response: The reference has been corrected, as per the BMC reference guide
Tables- please remove the visible vertical lines from your tables
Response: we have done so
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: we did not detect any errors in the manuscript.
Many thanks and best wishes